Author's response to reviews

Title: Canadian Emergency Department Triage and Acuity Scale: Implementation in a Tertiary Care Center in Saudi Arabia

Authors:

Naser B Elkum (nelkum@hotmail.com)
CarolAnne Barrett (doctorsaint1@hotmail.com)
Hisham Al-Omran (homran@kfshrc.edu.sa)

Version: 3 Date: 1 December 2010

Author's response to reviews:

Dear Rachel,

Please find attached a copy of the revised paper that we would like to be considered for publication in BMC Emergency Medicine as original research. The paper is entitled:

Canadian Emergency Department Triage and Acuity Scale: Implementation in a Tertiary Care Center in Saudi Arabia

The paper has been revised based on the reviewer’s comments and suggestions. We also are attaching our responses to both reviewers.

We did not identify any situation that might be perceived as a conflict of interest. Additionally, this manuscript has not been published nor submitted in whole or in part elsewhere.

We thought that the BMC Emergency Medicine would be an appropriate journal for this paper.

Looking forward to hearing from you.

Naser Elkum, PhD, P.Stat.